Author's response to reviews

Title: Association between pregnancy intention and optimal breastfeeding practices in the Philippines: a cross-sectional study

Authors:

Valerie Gilbert T. Ulep (vgtulep@gmail.com)
Maridel P Borja (dedelborja@yahoo.com)

Version: 3 Date: 13 August 2011

Author's response to reviews:

The title of tables 4 and 5 should be this:

Table 4. Result of the assessment of confounding for the association of pregnancy intention and breastfeeding initiation, by socio-economic status

Table 5: Result of the assessment of confounding for the association of pregnancy intention and breastfeeding duration, by socio-economic status